Author's response to reviews

Title: Implementation Challenges and Successes of a Population-based Colorectal Cancer Screening Program: A Qualitative Study of Stakeholder Perspectives

Authors:

Elizabeth G Liles (Elizabeth.G.Liles@kp.org)
Jennifer L Schneider (Jennifer.L.Schneider@kpchr.org)
Adrianne C Feldstein (Adrienne.C.Feldstein@kp.org)
David M Mosen (David.M.Mosen@kpchr.org)
Nancy Perrin (Nancy.Perrin@kpchr.org)
Ana G Rosales (Ana.G.Rosales@kpchr.org)
David H Smith (David.H.Smith@kpchr.org)

Version: 3 Date: 24 February 2015

Author's response to reviews: see over
February 23, 2015

Dear Dr. Sevdalis,

Thank you for the most recent communication about our paper (MS: 1131390102141809, “Implementation Challenges and Successes of a Population-based Colorectal Cancer Screening Program: A Qualitative Study of Stakeholder Perspectives”). We have read over the comments and address them below.

• “I'm still struggling to see how the issues raised in this study map onto the PRISM model, and the addition of Figure 2 hasn't helped me. Can the authors please provide a table along the lines of the one Feldstein and Glasgow 2008 have in their paper which places facilitators and barriers into the different sections of the PRISM model.”

We have provided this Table as Table 5 and have made reference to it in the Discussion section of the manuscript.

Thank you again for the thoughtful reviewer comments and for allowing us to address them. Please feel free to contact me with any additional questions or concerns.

Sincerely,

Elizabeth Liles, MD, MCR